Author's response to reviews

Title: Inflammatory Myofibroblastic Tumor Causing Unexplained Anemia in a Toddler: a case report. Mohammad Salameh1, MD, Iyad Sultan2, MD, Maha Barbar2, MD, Maysa Al Husieni3, MD, Abdullah Jameel4, MD, Khalil Ghandour1, MD, Ma'in Masarweh1, MD* 1. Department Of Surgery, King Hussein Cancer Center, Amman, Jordan Moh_s82@hotmail.com Jafapal48@yahoo.com 2. Department Of Pediatrics, King Hussein Cancer Center, Amman, Jordan isultan@khcc.jo mbarbar@khcc.jo 3. Department Of Pathology, King Hussein Cancer Center, Amman, Jordan mhussaini@khcc.jo 4. Department Of Radiology, King Hussein Cancer Center, Amman, Jordan Aj_nofal@hotmail.com *Correspondence To: Ma'in Masarweh, MD* Chief Section of Pediatric Surgery King Hussein Cancer Center, P O Box 2441 Amman, 11953, H.K. Jordan Tel +962795055920 Fax +96265342567 E mail: mmasarweh@khcc.jo

Authors:

Mohammad Salameh (moh_s82@hotmail.com)
Iyad Sultan (isultan@khcc.jo)
Maha Barbar (mbarbar@khcc.jo)
Maysa AlHussaini (mhussaini@khcc.jo)
Abdullah Jameel (aj_nofal@hotmail.com)
Khalil Ghandour (jafapal48@yahoo.com)
Ma'in Masarweh (mmasarweh@khcc.jo)

Version: 2 Date: 24 July 2010

Author's response to reviews:

Dear Editor:

In this revised manuscript you will find the changes required regarding the consent wording, authors contributions, and competing interest section.

Best wishes;

Ma'in Masarweh FRCSEd